Author’s response to reviews

Title: Health professional perceptions regarding screening tools for developmental surveillance for children in a multicultural part of Sydney, Australia

Authors:
Pankaj Garg (pankaj.garg@sswahs.nsw.gov.au)
My Trinh Ha (MyTrinh.Ha@sswahs.nsw.gov.au)
John Eastwood (John.eastwood@sswahs.nsw.gov.au)
Susan Harvey (sharv@bigpond.net.au)
Sue Woolfenden (Susan.Woolfenden@health.nsw.gov.au)
Elisabeth Murphy (lmurp@doh.health.nsw.gov.au)
Cheryl Dissanayake (c.dissanayake@latrobe.edu.au)
Katrina Williams (Katrina.Williams@rch.org.au)
Bin Jalaludin (b.jalaludin@unsw.edu.au)
Anne McKenzie (Anne.McKenzie@sswahs.nsw.gov.au)
Stewart Einfeld (Stewart.einfeld@sydney.edu.au)
Valsamma Eapen (v.eapen@unsw.edu.au)
Natalie Silove (Natalie.Silove@health.nsw.gov.au)
Kate Short (Kate.Short@sswahs.nsw.gov.au)

Version: 2 Date: 12 Sep 2017

Author’s response to reviews:

Authors’ response to Reviewer 2 (Frances Glascoe)

The authors have done a commendable job rewriting this paper - addressing all concerns reviewers raised. The manuscript is precise, clear and informative. I recommend publication.

We appreciate the comments and have not made any changes in the text of the manuscript.

Authors’ response to editorial office
1. In the Ethics approval and consent to participate section please clarify whether written informed consent was obtained from all participants.

A written informed consent was obtained from all participants. This is included in the revision 2 in the declaration section.

2. In the Consent for publication section please clarify if you obtained consent, written or verbal, from all participants for the publication of their anonymous quotes. If verbal, please state the reason and whether the ethics committee approved this procedure.

A written consent was obtained from all participants for the publication of their anonymous quotes. This is included in the revision 2 in the declaration section.

3. We note that the order of authors in the system is not the same as in the title page of the manuscript. Please ensure that the order of authors in the system and the title page of the manuscript are the same.

We have now checked that the order of the authors in the system and the title page of the manuscript are the same.

4. We note that the abbreviation CALD used in the manuscript text has not been included in the ‘List of Abbreviations’ section. Please ensure that all abbreviations present in the text are listed in this section.

We have included the missing abbreviation CALD in the ‘List of Abbreviations’ and have checked the all abbreviations present in the text are listed in this section.

5. Please include your title page as the first page of your manuscript instead of as a separate file.

The title page is included as the first page of the manuscript instead of as a separate file in the revised version. Please refer to the first page of revised version 2.
Authors’ response to editorial office

We note that the Declarations section in your manuscript has not been completed and may be missing information.

The following declaration section is included in the revised version 2 after the end of the main text in the manuscript document:

DECLARATIONS SECTION

• Ethics approval and consent to participate: Ethics approval was taken from the University of New South Wales and South Western Sydney Local Health District institutional committees. A written informed consent was obtained from all participants.

• Consent for publication: A written consent was obtained from all participants for the publication of their anonymous quotes.

• Availability of data and material: The datasets during and/or analyzed during the current study are available from the corresponding author on reasonable request as per the local ethics committee requirements.

• Competing interests: The authors declare that they have no competing interests.

• Funding: This study was funded by the National Health and Medical Research Council of Australia (APP 1013690), through a partnership grant with additional funding contribution from the NSW Ministry of Health and in-kind support from University of New South Wales, South Western Sydney Local Health District, Sydney Children’s Hospital Network, and La Trobe University. The funding organisations provided support for the personnel involved in the WMG study, but were not involved in the design, data collection, analysis and interpretation of the data.

Authors contributions: PG, JE, VE, SW, BJ, KW, AM, EM, CD, NS, SE, KS were involved in the initial design of this study. JE, PG and SH further refined the study. PG and SH collected the data, and PG, SH and MT analysed the data. PG and SH worked on the subsequent revisions and
all authors contributed to the intellectual content of the paper. All authors have read and approved the final version of this manuscript.

Acknowledgements

The authors thank Professor Margot Prior for her contribution to the development of the research proposal, colleagues from the child and family health nurses in the Liverpool/Fairfield/Bankstown areas and their managers Trish Clarke, Victoria Blight and Wendy Geddes, the staff of the postnatal wards at Liverpool and Bankstown hospitals, and the staff at the Clinical Information Department at Liverpool hospital. The authors also acknowledge the support of Deborah Beasley and April Deering for liaison with NSW Ministry of Health Department of Kids and Families, Alexandra Hendry (Research officer, Early Years Research Group) for her help in ethics submission and Pietty Greenwood and Kamalle Dadossey at Liverpool Resource Migrant Centre for their help in organising focus groups. The authors are grateful to all the interpreters and participants for their time and assistance in this research.

The ‘Watch Me Grow’ study group comprises of investigators Valsamma Eapen, Katrina Williams, Cheryl Dissanayake, Bin Jalaludin, Sue Woolfenden, John Eastwood, Stewart Einfeld, Elisabeth Murphy, Kate Short and Natalie Silove, as well as research associates Rudi Črnčec, Joseph Descallar, My Trinh Ha, Anne McKenzie, Stephen Matthey, and Amelia Walter and research staff Snehal Akre, Emma Axelsson, Annette Barone, Cherie Butler, Janelle Cleary, Mary Ha, Susan Harvey, Feroza Khan, Nicole Lees, Annabel Marsh, Van Nguyen, Laura Nichols, Bronwyn Overs, Ayesha Sadozai, Tara Shine, Olivia Wong, and Banosha Yakob.